Author’s response to reviews

Title: The endoscopic ultrasound probe findings in prediction of esophageal variceal recurrence after endoscopic variceal eradication therapies in cirrhotic patients: A cohort prospective study

Authors:

HUIGUO DING (dinghuiguo@medmail.com.cn)
Junfu zheng (zhengjunfu1979@foxmail.com)
Yuening Zhang (zhangyn-ty@163.com)
Shibing Zhang (zhangshibinyayy@163.com)
yue Li (liyue524@qq.com)
Lei Li (13699119545@163.com)
Huiguo Ding (dinghuiguo@ccmu.edu.cn)

Version: 2 Date: 05 Jan 2019

Author’s response to reviews:

Response letter

Jan 5, 2019

To editor and reviewers

Manuscript number: BMGE-D-18-00714

Title of manuscript: The endoscopic ultrasound probe findings in prediction of esophageal variceal recurrence after endoscopic variceal eradication therapies in cirrhotic patients: A cohort prospective study

Dear editor and reviewers,

On behalf of all co-authors, I would like to thank the reviewers and the editor for critical comments. With regard to the reviewer and editors’ comments on revising figure 1. In particular revise the lettering/labeling of all figures, which should be equal/in the same style ...

(e.g. Figure 3 red colored letters, figure 2 yellow colored letters, figure 1 yellow/green letters).
I have revised all figures in the same style.

Thanks!